Author’s response to reviews

Title: Rationale and methods of the European Study on Cardiovascular Risk Prevention and Management in Daily Practice (EURIKA)

Authors:

Fernando Rodríguez Artalejo (fernando.artalejo@uam.es)
Eliseo Guallar (eguallar@jhsphs.edu)
Claudio Borghi (claudio.borghi@unibo.it)
Jean Dallongeville (jean.dallongeville@pasteur-lille.fr)
Guy De Backer (guy.debacker@ugent.be)
Julian P Halcox (halcoxiu@cardiff.ac.uk)
Ramón Hernández-Vecino (Ramon.Hernandez@astrazeneca.com)
Francisco Javier Jiménez (javier.jimenez@astrazeneca.com)
Elvira L Massó-González (Elvira.MassoGonzalez@astrazeneca.com)
Joep Perk (joep.perk@lnu.se)
Philippe Gabriel Steg (gabriel.steg@bch.aphp.fr)
José R Banegas (joseramon.banegas@uam.es)

Version: 2 Date: 7 April 2010

Author’s response to reviews: see over
Dear Editor,

Please, find enclosed the manuscript entitled “Rationale and methods of the European Study on Cardiovascular Risk Prevention and Management in Daily Practice (EURIKA)”, to be considered for publication as a Study Protocol in BMC Public Health.

The EURIKA study is funded by AztraZeneca. The study protocol has been approved by the appropriate Ethics Committees in the 12 European participating countries. The letters of approval by one Ethics Committee in each country are submitted as additional files. Lastly, the EURIKA study has been registered in the Clinical Trials register, where it has been assigned the following number NCT00882336.

We inform you that all authors have contributed substantially to the manuscript. In particular, all of them have: 1) contributed to conception and design or analysis and interpretation of data, or both; 2) contributed to drafting of the manuscript or revising it critically for important intellectual content; and 3) provided final approval of the manuscript submitted.

We also inform you that: 1) the paper is not under consideration elsewhere, 2) none of the paper's contents have been previously published, 3) all authors have read and approved the manuscript, and 4) there is no potential conflict of interest regarding the publication of this manuscript.

We thank you for your attention in this matter.

Fernando Rodríguez-Artalejo, Eliseo Guallar, Claudio Borghi, Jean Dallongeville, Guy De Backer, Julian P. Halcox, Ramón Hernández-Vecino, Francisco Javier Jiménez, Elvira L. Massó-González, Joep Perk, Philippe Gabriel Steg, José R. Banegas, on behalf of EURIKA Investigators

Address for correspondence:
Fernando Rodríguez Artalejo, MD.
Department of Preventive Medicine and Public Health
School of Medicine
Universidad Autónoma de Madrid
C/Arzobispo Morcillo, sn
28029 Madrid (Spain)
Phone: +34914975444/3
Fax: +34914975353
E-mail: fernando.artalejo@uam.es